Author’s response to reviews

Title: Congenital Aniridia with Cataract: Case series

Authors:

Jin Wang (wjd3636@163.com)
Jing Zhang (lanbeisnow@163.com)
Ying Xiong (bear0629@163.com)
Jing Li (crystal_lijing@126.com)
Xiao Li (lxx0022@126.com)
Xue Liu (liuxue_1124@163.com)
Jing Zhao (cynthiazhao890401@163.com)
Frank F. Tsai (fftsai@gmail.com)
Jhanji Vishal (vishaljhanji@cunk.edu.hk)
Qi You (qishengbio@163.com)
Yao Huang (huangyao78@163.com)
Xiuhua Wan (xiuhuawan@126.com)

Version: 2 Date: 05 May 2017

Author’s response to reviews:

Dear editors:

Our team encountered difficulties in clinical trial registration matters. According to the requirements of WHO International Clinical Trials Registry Platform, we need to build a new type of raw database. It requires a lot of time, maybe a few months, to complete. The database also need to pass the audit and the publicity. We can get the TRN after all these work. We can provide our ethical approvals now. Cannot register it, or can you give us more time to complete the registration?
The other parts of the manuscript have been revised according to the requirement of reviewers and has been resubmitted. Please review. Thanks.

Thanks very much!

Kind regards!

Xiu Hua Wan